Author’s response to reviews

Title: Jingfukang induces anti-cancer activity through oxidative stress-mediated DNA damage in circulating human lung cancer cells

Authors:
Zujun Que (030110371@mail.ecust.edu.cn)
Zhiyi Zhou (zhouzhiyi77@hotmail.com)
Bin Luo (luo2014bin@163.com)
Changsheng Dong (dongchangsheng2010@163.com)
Yi Jiang (877708348@qq.com)
Hegen Li (shlaogen@163.com)
Jianhui Tian (hawk7150@hotmail.com)

Version: 2 Date: 06 Aug 2018

Author’s response to reviews:

Dear Editor,

We have tried our best to revise the article and try our best to reduce the duplication of the article. If our articles do not meet the requirements of your magazine, please let us know. Thank you!

Sincerely yours,